



Ministero del Lavoro
della Salute e
delle Politiche Sociali



**Ente Ospedaliero Specializzato in Gastroenterologia
"Saverio de Bellis"**

Istituto Nazionale di Ricovero e Cura a Carattere Scientifico
Ente di diritto pubblico D.M. del 31-3-1982
C.F. – P.IVA: 00565330727

via Turi, 27 70013 Castellana Grotte - Bari



Regione Puglia

ISTITUTO CERTIFICATO EN UNI ISO 9001:2008 n°IQ-1208-14



Laboratory of Nutritional Pathophysiology

April 02, 2014

Dear Editor,

as suggested, the running title has been added in the revised version of the manuscript.

Title: Probiotics against neoplastic transformation of gastric mucosa. Effects on cell proliferation and polyamine metabolism.

Author: Francesco Russo, Michele Linsalata, Antonella Orlando

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 6461-edited

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Dr. Francesco Russo, M.D.
Laboratory of Nutritional Pathophysiology
Scientific Institute for Digestive Diseases
IRCCS "Saverio de Bellis"
Via Turi 27
I-70013 Castellana Grotte (BA), Italy
Phone +390804994315



*Ministero del Lavoro
della Salute e
delle Politiche Sociali*



**Ente Ospedaliero Specializzato in Gastroenterologia
"Saverio de Bellis"**

**Istituto Nazionale di Ricovero e Cura a Carattere Scientifico
Ente di diritto pubblico D.M. del 31-3-1982**

C.F. – P.IVA: 00565330727

via Turi, 27 70013 Castellana Grotte - Bari



Regione Puglia

ISTITUTO CERTIFICATO EN UNI ISO 9001:2008 n°IQ-1208-14



Laboratory of Nutritional Pathophysiology

Fax +390804994313

e-mail francesco.russo@ircsdebellis.it